



A Preliminary Investigation of the Views of People With Parkinson's (With and Without Psychosis) and Caregivers on Participating in Clinical Trials During the Covid-19 Pandemic: An Online Survey

Katie McGoohan¹, Anneesa Amjad², Natasha Ratcliffe², Sagnik Bhattacharyya³, Gillian Granville², Matthew Sullivan², Lesley Gosden², Dag Aarsland^{1,4}, K. Ray Chaudhuri⁵, Dominic ffytche¹, Clive Ballard⁶ and Latha Velayudhan^{1,7*}

¹ Department of Old Age Psychiatry, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, United Kingdom, ² Parkinson's UK, London, United Kingdom, ³ Department of Psychosis Studies, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, United Kingdom, ⁴ Centre for Age-Related Medicine, Stavanger University Hospital, Stavanger, Norway, ⁵ Parkinson Foundation International Centre of Excellence, King's College Hospital and Kings College London, London, United Kingdom, ⁶ University of Exeter Medical School, St Luke's Campus, Exeter, United Kingdom, ⁷ South London and Maudsley National Health Service Foundation Trust, London, United Kingdom

Background: The coronavirus pandemic is having a profound impact on non-COVID-19 related research, including the delivery of clinical trials for patients with Parkinson's disease.

Objectives: A preliminary investigation to explore the views of Parkinson's disease (PD) patients, with and without experience of psychosis symptoms, and carers on the resumption of clinical research and adaptations to trials in light of COVID-19.

Methods: An anonymous self-administered online survey was completed by 30 PD patients and six family members/carers via the Parkinson's UK Research Support Network to explore current perceptions on taking part in PD research and how a planned clinical trial for psychosis in PD may be adapted so participants feel safe.

Results: Ninety-one percent of respondents were enthusiastic about the continuation of non-COVID-19 related research as long as certain safety measures were in place. Ninety-four percent stated that they would be happy to complete assessments virtually. However, they noted that care should be taken to ensure that this does not exclude participants, particularly those with more advanced PD who may require assistance using portable electronic devices. Regular and supportive communication from the research team was also seen as important for maintaining the psychological well-being of participants while taking part in the trial.

Conclusions: In the era of COVID-19 pandemic, standard approaches will have to be modified and rapid adoption of virtual assessments will be critical for the continuation

OPEN ACCESS

Edited by:

Gianfranco Spalletta, Santa Lucia Foundation (IRCCS), Italy

Reviewed by:

Nerisa Banaj, Santa Lucia Foundation (IRCCS), Italy Debanjan Banerjee, National Institute of Mental Health and Neurosciences (NIMHANS), India

> *Correspondence: Latha Velayudhan latha.velayudhan@kcl.ac.uk

> > Specialty section:

This article was submitted to Aging Psychiatry, a section of the journal Frontiers in Psychiatry

Received: 03 September 2020 Accepted: 24 November 2020 Published: 23 December 2020

Citation:

McGoohan K, Amjad A, Ratcliffe N, Bhattacharyya S, Granville G, Sullivan M, Gosden L, Aarsland D, Chaudhuri KR, ffytche D, Ballard C and Velayudhan L (2020) A Preliminary Investigation of the Views of People With Parkinson's (With and Without Psychosis) and Caregivers on Participating in Clinical Trials During the Covid-19 Pandemic: An Online Survey. Front. Psychiatry 11:602480. doi: 10.3389/fpsyt.2020.602480

December 2020 | Volume 11 | Article 602480

1

of clinical research. It is important that alongside the traditional methods, new tools are developed, and older ones validated for virtual assessments, to allow safe and comprehensive assessments vital for ongoing research in people with Parkinson's.

Keywords: hallucinations and delusions, patient and public involvement (PPI), COVID-19, clinical trial, psychosis, survey, Parkinson's disease

INTRODUCTION

The current coronavirus disease 2019 (COVID-19) pandemic is having a huge impact on healthcare systems and broader society across the world. Patients with chronic conditions are being significantly affected by loss of social contact, constraints on movement and disruption to access to both urgent and routine healthcare, with many outpatient appointments being canceled or postponed (1, 2). While implementation of telemedicine has increased dramatically (3), allowing for the continuation of ongoing care and remote monitoring (2, 4, 5) all of these factors raise some serious concerns for the health and well-being of patients with Parkinson's disease (PD) (6).

Although research has played an important part in our response to COVID-19, the outbreak has also had an impact on the conduct and delivery of non-COVID-19 related research. Most other research has been paused or significantly reduced, including many clinical trials for PD (7, 8). While this is understandable due to concern for the safety of trial participants and research staff, it is vital that trials resume in order to meet the unmet clinical needs for people with PD (9).

Parkinson's disease psychosis (PDP) refers to the range of hallucinations and delusions that occur in PD (10). Their prevalence increases with illness duration, with most patients eventually developing such symptoms (11). Although there are a number of clinical options available for treating PDP, they are either not very effective or require specialized monitoring of side effects (12). It is vitally important that promising interventions continue to be tested in the form of a clinical trial. In light of the current pandemic, changes to the traditional methods used for delivering research are therefore required so that research can resume as soon as it is safe to do so (13).

In considering these changes, it is vital that researchers work with patients and the public to understand their views on research participation during pandemic times and to ensure that trial adaptations are practical for future study participants. Patient and public involvement (PPI) is defined as research being carried out "with" or "by" members of the public rather than "to," "about," or "for" them (14). It is an essential activity in all stages of the research process and ensures the acceptability and relevance of research (15). This study reports the results of an online survey that was developed together with a group of PPI advisors to explore PD patients' views on taking part in research during the current pandemic and how a planned clinical trial for people with PDP may be best adapted so that participants feel safe. To ensure that the results of the survey were reflective of the participants who would be taking part in the planned clinical trial, we were specifically interested in recruiting people with experience of psychosis symptoms. However, as the results of this survey would be of interest to a much wider audience (e.g., those conducting research with patients with other neuropsychiatric disorders or older adults more generally), the survey was open to anyone with PD.

METHODS

Views were gathered using an online survey, which included a mixture of closed- and open-ended questions, the full details of which are provided in the Supplementary Material. The survey was co-created with three patient advisors (whose involvement was facilitated by Parkinson's UK) to ensure that the questions being asked in the survey were relevant to and informed by the perspective of people with PD. The patient advisors provided input on the questions, response options and format of the initial draft of the survey. The survey consisted of 19 questions: six questions gathering details about survey respondents, seven questions related to general perceptions about taking part in research at the current time, followed by five more specific questions about adaptations to the planned clinical trial investigating psychosis in PD. It should be noted that for some of the questions, multiple responses could be selected (see questions with "Tick all that apply" statement in Supplementary Material). The final question was a free text option asking what physical or psychological support would help people with PD take part in research at this time.

An invitation to participate was distributed by Parkinson's UK to their Research Support Network-an online network that brings together people driven to help find a cure and better treatments for PD (16). The survey was open to (inclusion criteria) anyone affected by Parkinson's- including partners, carers and family members of those with the condition, and people who had experience of hallucinations and/or delusions were especially encouraged to participate to ensure that any changes being made to the planned clinical trial were inclusive and accessible to future participants. The questionnaire was in English and sent out via email, so excluded those who were not-fluent in English or who did not have access to computers or have an email address. Caregivers or family members were able to complete the survey on behalf of a person affected by PD (e.g., the person with PD was unable to complete the survey themselves). People expressed their interest to Parkinson's UK and were then sent a link to the survey, along with a plain English summary of the trial. The survey was administered using SmartSurvey, an online survey software and questionnaire tool. As any adaptations to the planned clinical trial were required to be processed within a timely manner, the survey was open to responses from 26 June to 6 July 2020.

Survey responses were fully anonymized and no identifiable information was collected. The survey was conducted as a PPI activity therefore no ethical approval was required. Informed consent was obtained from all participants.

Simple statistical summaries were generated for the closed form responses to each survey item. Since responses to the open-ended questions were fairly succinct, no formal qualitative analysis or prespecified framework were imposed on the open response data. To aid interpretation of the quantitative results, free-text responses were grouped into categories and classified as being positive, negative or neutral in tone (for example, whether respondents were generally positive, had reservations or were reluctant about the continuation of research in their comments).

RESULTS

Thirty people with PD (83%) and six carers, partners or family members (17%) completed the survey. Fourteen (47%) of the respondents with PD had experience of psychosis symptoms: five had experience of hallucinations, two of delusions, and seven of both symptoms of psychosis. All six carers lived with someone who had experience of psychosis symptoms: three each with experience of hallucinations alone and both hallucinations and delusions. **Table 1** details characteristics of the survey respondents and **Table 2** the main results of the closed form responses.

General Feelings on Taking Part in Research

When asked what their feelings were about taking part in research at present, or in the near future, given the current COVID-19 pandemic, the majority of respondents (69%) were positive about the continuation of non-COVID-19 related research as long as it was safe to do so. This was supported by numerous comments about the importance of research:

"It is important to carry on with research as life goes on research is still necessary to help with finding a cure. Without this there will not be any answers. Whatever we come up against we all have to deal with."

"I am keen to take part in any trials regarding PD. There is no cure, that doesn't mean we shouldn't be looking for one."

"I think research is very important and if it can be carried out in a safe way I'm happy to participate."

"I am quite happy to take part in research if it may help my wife and the many others suffering from Parkinson's."

Some respondents (22%) were positive but had reservations or requirements for participation to feel safe:

"I have no objection to taking part in clinical research, providing my participation takes place within the safe rules governing COVID-19."

"I have to take sensible precautions."

"I would like to be tested to see if I have had COVID-19 before taking part."

		PD participants, n = 30 (%)	Carers, <i>n</i> = 6 (%)
Time since PD diagnosis	Within the last year	1 (3)	0 (0)
	1–5 years	12 (40)	1 (17)
	5–10 years	8 (27)	2 (33)
	10–15 years	3 (10)	2 (33)
	More than 15 years	6 (20)	1 (17)
Experience of psychotic symptoms*	Experience of hallucinations and delusions	7 (24)	3 (50)
	Experience of hallucinations	5 (17)	3 (50)
	Experience of delusions	2 (7)	O (O)
	No experience of hallucinations or delusions	15 (52)	O (O)
Living arrangements	Living with partner	25 (83)	5 (83)
	Living with family/friends	1 (3)	O (O)
	Living on their own	4 (13)	O (O)
	Other (live in carer)	O (O)	1 (17)
Previous participation in research	No previous participation	12 (40)	3 (50)
	Participation in online research	17 (57)	2 (33)
	Participation in a clinical trial	3 (10)	1 (17)
	Other	1 (3)	O (O)

*Missing data from one participant with PD. PD. Parkinson's Disease. TABLE 2 | Survey responses to closed form questions from PD participants with and without experience of psychosis symptoms and from carers, family members or partners.

Survey questions	Response options	PD participant with psychosis symptoms, n = 14* (%)	PD participant without psychosis symptoms, $N = 15^*$ (%)	Carers, family members or partners, n = 6 (%)
Preference for study visit location	More comfortable taking part in research from home	4 (29)	4 (27)	5 (83)
	More comfortable taking part in research that involved a visit to a clinical setting	2 (14)	1 (7)	0 (0)
	Comfortable either way	8 (57)	10 (67)	1 (17)
	Not comfortable either way	O (O)	O (O)	O (O)
What might help a home	PPE for the researcher	9 (64)	11 (73)	4 (67)
visit from a researcher feel	PPE for participant	8 (57)	8 (53)	1 (17)
safe	The researcher traveling by car (not using public transport)	8 (57)	9 (60)	4 (67)
	The researcher having regular tests for COVID-19	8 (57)	9 (60)	5 (83)
Maximum length of time for	1 h	1 (7)	7 (47)	3 (50)
home visit	2 h	11 (79)	5 (33)	3 (50)
	3h	1 (7)	1 (7)	0 (0)
	4 h	0 (0)	0 (0)	0 (0)
	5 h	1 (7)	2 (13)	0 (0)
What might help a study visit in a clinical setting feel safe	PPE for the researcher	10 (71)	10 (67)	4 (67)
	Participants being required to wear a mask	9 (64)	10 (67)	3 (50)
	Participants being required to use their own personal transport or being offered a taxi	12 (86)	9 (60)	4 (67)
	Thorough cleaning of assessment rooms in between participants	9 (64)	8 (53)	3 (50)
Maximum length of time for visit to clinical setting	1 h	O (O)	5 (33)	3 (50)
	2 h	12 (86)	6 (40)	3 (50)
	3h	0 (0)	1 (7)	O (O)
	4 h	1 (7)	O (O)	O (O)
	5 h	1 (7)	3 (20)	O (O)
Willingness to complete	Yes	13 (93)	15 (100)	5 (83)
study assessments virtually	No	1 (7)	O (O)	1 (17)
Willingness to take a finger-prick blood test at home	Yes	14 (100)	14 (93)	6 (100)
	Not sure	O (O)	1 (7)	0 (0)
Willingness to take a	Yes	2 (14)	2 (13)	1 (17)
pregnancy test at home	No	2 (6)	O (O)	O (O)
	Not applicable	10 (71)	13 (87)	5 (83)
Willingness to track of study	Yes	14 (100)	14 (93)	5 (83)
drug compliance at home	Not sure	0 (0)	1 (7)	1 (17)

*Missing data from one participant with PD who did not respond to the question asking whether they had experience of hallucinations and/or delusions.

PPE, Personal protective equipment.

It was also clear from some of the comments that participants were particularly reluctant to attend hospital visits, preferring research activities to be conducted virtually: "I feel comfortable with taking part in PD related research activities but would prefer not to have to visit hospitals and clinics."

"I am happy to do on-line research but will not attend either a hospital or a face to face meeting of any type."

Only a couple of respondents (6%), both of whom had experience of psychosis symptoms, expressed reluctance to participate in research at the present time.

Preferred Location and Length of Time for Visits

When asked whether respondents would feel more comfortable taking part in research if they did not have to visit a clinical setting, over 50% said they would feel comfortable either way, 39% said they would feel more comfortable taking part in research from home, 8% said they would feel more comfortable taking part in research that involved a clinical setting and no respondents selected "I would not feel comfortable either way." Carers were more inclined (83%) toward study visits taking place at their home compared to PD participants (27%). These respondents were often caring for people with complex needs and did not think it appropriate to visit other settings.

In response to the question asking how they would feel about a researcher visiting their home to conduct study visits, 69% of respondents expressed that they would be agreeable to a researcher conducting a home visit, and a further 22% stated that they would be willing as long as appropriate safety measures were taken. This was backed up by open-text comments throughout the survey:

"I am happy to do that. As long as all precautionary measures are put in place on both sides."

"As long as exemplary COVID-19 precautions are taken—not a problem for me."

To help make a home visit from a researcher feel safe, many respondents selected that they would like to see all the multiplechoice options that we provided in the survey being applied: (1) personal protective equipment (PPE) for the researcher (67%), (2) the researcher having regular tests for COVID-19 (64%), (3) the researcher traveling by car (not using public transport) (61%), and (4) PPE for the participant (47%). Further suggestions included meetings to be held outdoors (for example, in participant's gardens) and maintaining a safe distance between the participant and researcher.

Similarly, when asked what would help to make a study visit at a local hospital feel safe, most respondents selected all of the multiple-choice options that we provided in the survey: (1) participants being required to use their own personal transport or being offered a taxi (72%), (2) researchers wearing personal protective equipment (67%), (3) participants being required to wear a mask (61%), and (4) thorough cleaning of assessment rooms in between participants (58%). In addition, most respondents were willing for both home (67%) and hospital (75%) visits to last 2 hours or more.

Only 17% of carers and 44% of PD participants with experience of delusions felt that participants should be required to wear PPE in their own homes, compared to 60% of PD participants without symptoms of delusions (no experience of psychosis or symptoms of hallucinations only). PD participants with experience of delusions were also more likely to require researchers (89%) and participants (100%) to use personal transport when traveling to study visits compared to participants without these symptoms (45 and 60% respectively).

Virtual Assessments

A breakdown of the types of assessment that respondents would be amenable to completing virtually as part of the planned clinical trial is provided in **Table 3**. Only two respondents, both of whom had experience of psychosis, said they would not feel comfortable carrying out study assessments virtually.

Telephone calls was the preferred method for remote followup compared to video call or online surveys. Reasons for this varied in the open text comments but were mainly driven by issues with connectivity and feelings of discomfort toward the use of internet-based technology:

"On the whole I don't like video calls as I feel self-conscious, but I would be prepared to overcome this."

"I am not comfortable using my computer, so I prefer to fill in a real form than a virtual one."

When asked what would make virtual assessments easier to complete, respondents noted that questionnaires should be easy to understand and come with clear guidance and instructions. Support from a family member or carer and flexibility in the way that assessments could be completed (i.e., ability to complete in several sittings), were also seen as important in the comments:

"My mother would need assistance of a carer to complete any assessment and responses would be by 3rd party from the carers."

Almost all respondents were willing to take a finger-prick home blood test (as a remote alternative to venepuncture) (97%) as long as they had clear instructions and, if required, the support of a carer. Respondents were also willing to monitor compliance with study drugs (i.e., pill counts (94%)) at home. Suggestions for what would make this easier to complete included the use of an online diary and pill dispensers.

Additional Support

The final question in the survey asked what other physical or psychological support would help people with PD take part in a research study at this time. Supportive and regular contact from research teams, preferably with a designated contact for the length of the study, was frequently suggested by respondents. For those with psychosis symptoms, importance was also given in the comments to the emotional support required when answering difficult questions relating to their symptoms:

"Dealing with distress caused by recalling upsetting episodes of hallucinations."

Other recommendations included maintained engagement with the progress of the trial and the requirement for peer or carer support to assist with participation in the trial, particularly when considering remote assessments that require use of technology. TABLE 3 | Study assessments that respondents would be willing to complete virtually or whether they would prefer face to face.

	Telephone, <i>N</i> = 36 (%)	Video call, <i>n</i> = 36 (%)	Online survey, n = 36 (%)	Prefer face to face, n = 36 (%)
Consent	32 (89)	24 (67)	24 (67)	3 (8)
Medical history and current medications	30 (83)	23 (64)	24 (67)	5 (14)
Sociodemographic information	31 (86)	22 (61)	25 (69)	3 (8)
Adverse events	27 (75)	22 (61)	21 (58)	7 (19)
Motor symptoms of PD	28 (78)	24 (67)	24 (67)	4 (11)
Non-motor symptoms of PD	27 (75)	23 (64)	26 (72)	5 (14)
Quality of life	29 (81)	21 (58)	24 (67)	6 (17)

PD, Parkinson's Disease.

DISCUSSION

Following the Guidance for Reporting Involvement of Patients and Public, 2 (GRIPP2; see **Supplementary Material**), this paper highlights the importance of PPI in ensuring that any changes that are made to the way that we deliver research for people with PD in a world with COVID-19 are acceptable to those who will be participating (17).

Our survey findings suggest that despite the current pandemic, PD patients and their carers see the importance of research and remain enthusiastic about participation. Although a small proportion of respondents in this survey was anxious about taking part in research in the short term, most respondents were comfortable for researchers to conduct face to face study assessments, as long as adequate safety precautions were in place.

Much of the open-ended data provided by the respondents emphasized the importance of taking a flexible approach to research. Following the COVID-19 pandemic, there has been rapid growth and development in the area of telemedicine and digital healthcare, which can be easily adapted for use in PD research for the evaluation of both motor and non-motor symptoms (2). While virtual follow-up may not be possible for all types of assessment [for example, physical examinations (18)], it may help reduce the length of time required for in-person visits, thereby potentially reducing the likelihood of infection acquired during the study visit (3, 19). Whilst almost all respondents reported that they would feel comfortable carrying out some of the study assessments and questionnaires virtually, care would be needed to ensure that we do not exclude participants, particularly those without the skills or support to use portable electronic devices (20). Researchers should therefore be prepared to provide additional information and one-to-one support to ensure that participants feel comfortable. Providing options in the way that participants can interact with researchers during visits and throughout the study was identified as being of the utmost importance so that participants felt supported. Virtual assessments in the clinical trial may therefore be used as part of a flexible package of follow-up methods, alongside traditional telephone and face to face approaches (21).

Although the sample size of our survey was small, this is likely attributable to our short period of data collection and request for people who had experience of psychosis. Nevertheless, this was important to ensure that some of the views were reflective of the sample who will be taking part in the planned clinical trial for psychosis in PD. Psychosis typically occurs in the later stages of PD, with risk factors including older age, increased duration and severity of PD, and significant psychiatric or medical comorbidity (11, 22). While there is currently no evidence to suggest that those with PD are at increased risk of contracting coronavirus (9, 23), advanced PD patients [for whom the prevalence of PDP ranges from 20 to 70% (11)] may be more susceptible and at greater risk for respiratory complications after a COVID-19 infection (24). Despite this, attitudes toward the continuation of research in the current climate were equally positive among participants both with, and without, experience of psychosis. Analysis between the two groups showed that those with experience of PDP (particularly carers and family members) were more inclined toward face to face visits taking place in a home setting, and open responses emphasized the requirement for ongoing physical and psychological support from carers and the research team, however, the sample size is too small to draw conclusions about such patterns.

All carers, partners and family members who participated in the survey lived with someone who had experience of psychosis symptoms. It is therefore not surprising that carers preferred study visits to take place in an environment in which the participant is familiar. Although our numbers were too small to demonstrate differences between the groups with any certainty, it does seem that carers and participants with experience of delusions were also less likely to suggest that participants wear PPE during a home visit. While the benefits of PPE are clear, we must be aware of the potentially disorientating impact that wearing PPE may have on patients, particularly those suffering from symptoms of psychosis. It is therefore important that researchers provide participants with additional information on what to expect during study participation, including clear information about any COVID safety measures that are in place, prior to the study visit. Not only will this provide reassurance, but it may also help reduce any anxiety and enable a more informed decision about research participation.

In the early stages of PD, hallucinations typically occur with insight initially preserved, whereas in the later disease stages, patients might not recognize the hallucinations as unreal anymore due to the onset of false beliefs (delusions) (10, 12). Given this progression, one might anticipate respondents with experience of delusions to have impaired insight compared to those with no experience of psychosis symptoms or experience of hallucinations only. Interestingly however, this did not appear to be the case among respondents of this survey. The only difference in responses between PD participants with and without symptoms of delusions was the requirement for researchers and participants to use their own personal transport or taxis when traveling to study visits, perhaps showing an increased awareness of the risks associated with COVID-19 among this more advanced PD population group. However, such an interpretation is tentative because of the small size of the sub-groups.

Limitations

Only 11% of participants had previous experience of a clinical trial, so may not be familiar with the types of procedures and assessments that this would typically involve. However, the fact that these participants still expressed their interest in taking part in future trials and research is noteworthy. It is also worth mentioning that self-report of psychotic symptoms would depend on respondent's insight and willingness to share such information, hence could have been under-reported. The results of this survey should also be considered in the context of a group with a specific interest in PD research. Despite this, it was clear from the responses that members of the Research Support Network used their knowledge of other people with PD as well as their own experiences, and their views were invaluable in informing how best to adapt the clinical trial. PPI members will continue to inform and improve participation in this research through their involvement with the trial's Patient Advisory Group and Trial Steering Committee.

Although our sample size was small, this survey was conducted with a fairly unique population, and highlights for the first time the views of PD patients and caregivers on taking part in research during the COVID-19 pandemic. The sample size was also deemed appropriate because of the exploratory nature of this research and the focus on identifying initial attitudes about the topic. While this study focused on patients with PD, many of the broader insights and recommendations will also be applicable to those involved in the design and delivery of clinical research with patients with other neuropsychiatric disorders as well an older adult population more generally.

Future Research

Although the sample size was deemed appropriate due to the exploratory nature of this research, a larger sample may have identified additional viewpoints or provided more nuanced explanations for PD patients and caregivers attitudes toward the continuation of research during pandemic times. It would therefore be useful to conduct a broader survey on the topic of adapting research in light of COVID-19, helping ensure that adaptations made to clinical studies are informed by the needs of people affected by Parkinson's. Future research would benefit from collecting data on current neurological, psychiatric and cognitive status, treatment and caregiver burden to determine whether these symptoms would influence responses. Researchers might also consider asking questions about all the types of devices participants have access to in their home and/or

their technical literacy. It would be useful for upcoming clinical trials in PD to add a qualitative component to their study to capture the views and opinions of participants who are taking part in research on any ethical challenges or other barriers encountered, especially during the period of the pandemic.

CONCLUSION

Non-COVID-19 related research remains of critical importance and must not be neglected. Although patients with PD remain enthusiastic about participation in research, a flexible approach to the way that we redesign and deliver our studies in a world where travel and face to face contact are restricted is required. New tools should be developed and existing tools should be validated for virtual use as a matter of urgency to ensure research can continue to be delivered during the COVID-19 pandemic.

DATA AVAILABILITY STATEMENT

The raw data supporting the conclusions of this article will be made available by the authors, without undue reservation.

ETHICS STATEMENT

Ethical review and approval was not required for the study on human participants in accordance with the local legislation and institutional requirements. The patients/participants provided their written informed consent to participate in this study.

AUTHOR CONTRIBUTIONS

LV, SB, KM, NR, and AA: conceptualization. KM, AA, NR, LG, MS, GG, LV, and SB: survey design. AA and KM: analysis and interpretation of data. KM, LV, AA, NR, GG, and MS: writing—original draft. KM, AA, NR, MS, GG, LV, DA, KC, SB, DF, and CB: writing—revising and providing the final approval of work. All authors contributed to the article and approved the submitted version.

FUNDING

The authors received no specific funding pertaining to this work. SB was supported by grants from the National Institute of Health Research (NIHR) Efficacy and Mechanism Evaluation scheme (UK) and SB, LV, DA, KC, CB, and DF are in receipt of funding from Parkinson's UK (G-1901). AA and NR are employed by Parkinson's UK. This paper represents independent research partly funded by the National Institute for Health Research (NIHR) Biomedical Research Center at South London and Maudsley NHS Foundation Trust and King's College London. The views expressed are those of the authors and not necessarily those of the NHS, the NIHR, PUK, or the Department of Health and Social Care.

ACKNOWLEDGMENTS

The authors are grateful to the members of the Parkinson's UK Research Support Network who dedicated their time to participate in our survey.

REFERENCES

- Helmich RC, Bloem BR. The impact of the COVID-19 pandemic on Parkinson's disease: hidden sorrows and emerging opportunities. *J Parkinsons Dis*. (2020) 10:351–4. doi: 10.3233/JPD-202038
- Miele G, Straccia G, Moccia M, Leocani L, Tedeschi G, Bonavita S, et al. Telemedicine in Parkinson's disease: how to ensure patient needs and continuity of care at the time of COVID-19 pandemic. *Telemedicine e-Health.* (2020). doi: 10.1089/tmj.2020.0184. [Epub ahead of print].
- Dorsey ER, Okun MS, Bloem BR. Care, convenience, comfort, confidentiality, and contagion: the 5 C's that will shape the future of telemedicine. *J Parkinsons Dis.* (2020) 10:893–7. doi: 10.3233/JPD-202109
- Cubo E, Hassan A, Bloem BR, Mari Z, MDS-Telemedicine Study Group. Implementation of telemedicine for urgent and ongoing healthcare for patients with Parkinson's disease during the COVID-19 pandemic: new expectations for the future. *J Parkinsons Dis.* (2020) 10:911– 3. doi: 10.3233/JPD-202108
- Motolese F, Magliozzi A, Puttini F, Rossi M, Capone F, Karlinski K, et al. Parkinson's disease remote patient monitoring during the COVID-19 lockdown. *Front Neurol.* (2020) 11:567413. doi: 10.3389/fneur.2020.567413
- Shalash A, Roushdy T, Essam M, Fathy M, Dawood NL, Abushady EM, et al. Mental health, physical activity, and quality of life in Parkinson's disease during COVID-19 Pandemic. *Mov Disord.* (2020) 35:1097-9. doi: 10.1002/mds.28134
- World Health Organization. COVID 19 strategy update- 14 April 2020. (2020). Available online at: https://www.who.int/publications/i/item/covid-19-strategy-update---14-april-2020 (accessed 15th May, 2020).
- Jacqui J. Clinical trials suspended in UK to prioritise covid-19 studies and free up staff. *BMJ*. (2020) 368:m1172. doi: 10.1136/bmj.m1172
- Papa SM, Brundin P, Fung VSC, Kang UJ, Burn DJ, Colosimo C, et al. Impact of the COVID-19 pandemic on Parkinson's disease and movement disorders. *Mov Disord*. (2020) 35:711–5. doi: 10.1002/mds.28067
- ffytche DH, Creese B, Politis M, Chaudhuri KR, Weintraub D, Ballard C, et al. The psychosis spectrum in Parkinson disease. *Nat Rev Neurol.* (2017) 13:81–95. doi: 10.1038/nrneurol.2016.200
- Levin J, Hasan A, Höglinger GU. Psychosis in Parkinson's disease: identification, prevention and treatment. J Neural Transmission. (2016) 123:45–50. doi: 10.1007/s00702-015-1400-x
- Taddei RN, Cankaya S, Dhaliwal S, Chaudhuri KR. Management of psychosis in Parkinson's disease: emphasizing clinical subtypes and pathophysiological mechanisms of the condition. *Parkinsons Dis.* (2017) 2017:3256542. doi: 10.1155/2017/3256542
- MHRA. Managing clinical trials during Coronavirus (COVID 19). (2020). Available online at: https://www.gov.uk/guidance/managing-clinical-trialsduring-coronavirus-covid-19 (accessed 16 July, 2020).
- INVOLVE. What is public involvement in research? (2020). Available online at: http://www.invo.org.uk/find-out-more/what-is-public-involvement-inresearch-2/ (accessed 16 July, 2020).

SUPPLEMENTARY MATERIAL

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fpsyt. 2020.602480/full#supplementary-material

- National Institute for Health Research. Patient and public involvement in health and social care research: a handbook for researchers. (2014). Available online at: https://oxfordbrc.nihr.ac.uk/wp-content/uploads/2017/ 03/RDS-PPI-Handbook-2014-v8-FINAL-2.pdf (accessed 20 July, 2020).
- Parkinson's UK. *Get Involved in Research.* (2020). Available online at: https://www.parkinsons.org.uk/research/get-involved-research (accessed 20 July, 2020).
- Staniszewska S, Brett J, Simera I, Seers K, Mockford C, Goodlad S, et al. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. *BMJ*. (2017) 358:j3453. doi: 10.1136/bmj.j3453
- Grossman SN, Han SC, Balcer LJ, Kurzweil A, Weinberg H, Galetta SL, et al. Rapid implementation of virtual neurology in response to the COVID-19 pandemic. *Neurology*. (2020) 94:1077–87. doi: 10.1212/WNL.0000000000 09677
- Ben-Pazi H, Browne P, Chan P, Cubo E, Guttman A, Hassan A, et al. The promise of telemedicine for movement disorders: an interdisciplinary approach. *Curr Neurol Neurosci Rep.* (2018) 18:26. doi: 10.1007/s11910-018-0834-6
- Foster MV, Sethares KA. Facilitators and barriers to the adoption of telehealth in older adults: an integrative review. *Comput Inform Nurs.* (2014) 32:523– 33. doi: 10.1097/CIN.00000000000105
- Richardson SJ, Carroll CB, Close J, Gordon AL, O'Brien, J, Quinn, TJ, et al. Research with older people in a world with COVID-19: identification of current and future priorities, challenges and opportunities. *Age Ageing*. (2020) 49:901–6. doi: 10.1093/ageing/afaa149
- 22. Sawada H, Oeda T, Yamamoto K, Umemura A, Tomita S, Hayashi R, et al. Trigger medications and patient-related risk factors for Parkinson disease psychosis requiring anti-psychotic drugs: a retrospective cohort study. *BMC Neurol.* (2013) 13:145. doi: 10.1186/1471-2377-13-145
- Stoessl AJ, Bhatia KP, Merello M. Movement Disorders in the World of COVID-19. *Mov Disord.* (2020) 35:709–10. doi: 10.1002/mds. 28069
- Antonini A, Leta V, Teo J, Chaudhuri KR. Outcome of Parkinson's disease patients affected by COVID-19. *Mov Disord.* (2020) 35:905– 8. doi: 10.1002/mds.28104

Conflict of Interest: The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Copyright © 2020 McGoohan, Amjad, Ratcliffe, Bhattacharyya, Granville, Sullivan, Gosden, Aarsland, Chaudhuri, ffytche, Ballard and Velayudhan. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.